OODS Tutorial 4

© 1999/2000 Michael Golm Universität Erlangen Nürnberg, Institut für Informatik 4

Homework Assignment #3: Drawing Editor Save / Load 0 points

Duringthis assignment you willimprovethe DrawingEditoryou programmed in the previous assignment. Copy all filesof assignment #2 in the assignment3 directory.Youcan also use the example solution that is available in/proj/i4oods/pub/assignment2.

a) Adda method void resize(int w, int h) to the Shape class. Additionalto the position of the shape the shape dimension should be given on the command line as follows:

java DrawingEditor Rect 120 80 20 30 Circle 50 150 60 50

The resize method should be called to set the shape’sdimension.Please ignorethe fact that a circlenow could be oval. Duringthe creation of the class objects and shape instances exceptions couldbe thrown.React to these exceptions by printingan error message and terminating the program.

b) The shape specificationsthat were read fromthe command line parameters should now be read froma fileifa filenameis givenas a command lineparameter, e.g. java DrawingEd-

itor mydrawing.drw. The fileshould contain one shape per line.Asample inputfileis
/proj/i4oods/pub/assignment3/drawing.drw. Ifno filename is given, the edi-

tor should start up with no shapes in the drawing. To load a drawingyou should implementa load() method in the DrawingEditor. This method shoud throwan IOException ifsomethinggoes wrongreading the fileand a For-

matException when something goes wrong creating the class objects and Shape in-

stances orconvertingthe position/dimensiontoint. Youshoulddefinethe FormatExcep-

tion class. IOException can be importedfromjava.io. Ifone ofthese exceptions are

thrown when the drawing is loaded, the program should terminate. Because class names may containunicode characters youshould use a Reader to read the drawing file. Anappletcan notaccess localfiles.Thereforeyouhave torunyourprogramas an application.